Author's response to reviews

Title: Facial skin metastasis due to small cell lung cancer. A case report.

Authors:

Nikolaos Barbetakis (nibarb@otenet.gr)
Georgios Samanidis (gsaman@yahoo.gr)
Dimitrios Paliouras (dpaliour@yahoo.gr)
Elpida Samanidou (elsaman@yahoo.gr)
Zoi Tzimorota (zoitzimor@yahoo.gr)
Christos Asteriou (chasteriou@yahoo.gr)
Persefoni Xirou (persxirou@yahoo.gr)
Christodoulos Tsilikas (ctsilikas@yahoo.gr)

Version: 4 Date: 28 September 2008

Author's response to reviews:

Thessaloniki 28/9/2008
With respect to the associate editor's comments, this is a point by point response.

1. Comments on the special stains and differential diagnosis have been removed in the discussion.
2. We have tried to improve the quality of submitted photos.
3. We have corrected grammatical and syntax errors.

Thank you for your help.

N. Barbetakis